



Kailei Tang, Zhiyan Dong \*, Wenxiang Shi and Zhongxue Gan

The Academy for Engineering and Technology, Fudan University, Shanghai 200433, China; 20210860147@fudan.edu.cn (K.T.)

\* Correspondence: dongzhiyan@fudan.edu.cn; Tel.: +86-189-4363-1070

Abstract: As the Internet of Things devices are deployed on a large scale, location-based services are being increasingly utilized. Among these services, kNN (k-nearest neighbor) queries based on road network constraints have gained importance. This study focuses on the CkNN (continuous k-nearest neighbor) queries for non-uniformly distributed moving objects with large-scale dynamic road network constraints, where CkNN objects are continuously and periodically queried based on their motion evolution. The present CkNN high-concurrency query under the constraints of a super-large road network faces problems, such as high computational cost and low query efficiency. The aim of this study is to ensure high concurrency nearest neighbor query requests while shortening the query response time and reducing global computation costs. To address this issue, we propose the DVTG-Index (Dynamic V-Tree Double-Layer Grid Index), which intelligently adjusts the index granularity by continuously merging and splitting subgraphs as the objects move, thereby filtering unnecessary vertices. Based on DVTG-Index, we further propose the DVTG-CkNN algorithm to calculate the initial kNN query and utilize the existing results to speed up the CkNN query. Finally, extensive experiments on real road networks confirm the superior performance of our proposed method, which has significant practical applications in large-scale dynamic road network constraints with non-uniformly distributed moving objects.

Keywords: k-nearest neighbor query; moving object; road network

# 1. Introduction

Location-based services have become increasingly important in emergencies, such as COVID-19 or flood disasters, where residents may be confined to their homes or communities [1]. Such services often involve location-based queries, with *Ck*NN (continuous k-nearest neighbor) queries being among the most important [2]. The *Ck*NN problem in the road network environment is particularly challenging, but socially and commercially valuable [3–5]. For example, the Meituan errand service sends information to several riders closest to the user [6]. In addition, *Ck*NN query technology has also promoted the improvement of mobile taxi software, such as Didi and Uber, which can send taxi requests to vehicles close to users [7].

However, several factors make this problem highly challenging. First, the query object dynamically moves within a large-scale road network, which implies that the query response time of urban services is affected by the fast-moving objects [8]. This is further compounded by the fact that objects are continuously on the move [9], thus making the indexing of moment-to-moment moving objects on the road network a major challenge. Secondly, there is a substantial volume of concurrent query requests [10]. For instance, in Beijing, millions of queries are generated every day, and at peak times, tens of thousands of queries per second are processed. Consequently, the need to process high-concurrency queries to identify kNN moving objects poses yet another challenge [11]. Furthermore, the existing research scarcely delves into the impact of the distribution of moving objects



Citation: Tang, K.; Dong, Z.; Shi, W.; Gan, Z. A Dynamic Grid Index for *Ck*NN Queries on Large-Scale Road Networks with Moving Objects. *Appl. Sci.* 2023, *13*, 4946. https://doi.org/ 10.3390/app13084946

Academic Editor: Donghai Guan

Received: 21 March 2023 Revised: 10 April 2023 Accepted: 11 April 2023 Published: 14 April 2023



**Copyright:** © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). in the road network on the performance of neighbor queries. In practical scenarios, the distribution of moving objects in urban environments tends to follow a Zipf distribution; yet, most of the existing studies have not effectively optimized this situation [12].

To address these challenges, this paper presents a novel index, the DVTG-Index (Dynamic V-Tree Double-Layer Grid Index), and a new algorithm, the DVTG-CkNN. The DVTG-Index introduces two new vertex types, supports dynamic updates of moving objects, facilitates the allocation of computing resources, and facilitates efficient kNN queries. The index initially partitions the entire graph recursively into several grid subgraphs based on the quadtree structure, and subsequently constructs a new grid index to manage these subgraphs. Active vertices associate moving objects with their corresponding network vertices on the road network, and boundary vertices enable the querying of the distance from a point to other vertices in the grid, enhancing computational efficiency.

Our contributions are as follows:

(1) We introduce the DVTG-Index, a scalable, efficient, and adaptable index that accommodates road network datasets and allows for adaptive grid merging and splitting for different regions to obtain the appropriate index granularity and low maintenance cost.

(2) We propose a robust update strategy that supports the dynamic update of moving objects by associating them with vertices to achieve real-time updates of moving objects. This approach better supports the handling of large mobile objects on large-scale road networks for mobile updates.

(3) Building on the DVTG-Index, we design a novel *k*NN search method that leverages the index's capabilities to compute *k*-nearest objects.

(4) We conduct extensive experiments on four real datasets to evaluate our approach's efficacy. Our experimental results demonstrate that our approach significantly outperforms existing baseline methods.

This paper is structured as follows: Section 1 presents an overview of the problem, while Section 2 reviews the relevant literature. Section 3 introduces the method overview and relevant definitions in this work. Section 4 details the DVTG-Index's structure and Section 5 outlines an efficient *k*NN search algorithm. Section 6 presents our experimental results. Section 7 concludes the paper.

# 2. Related Work

The *Ck*NN query has been an extensively studied research topic, with numerous theories and algorithms developed to address it. The research at present aims to reduce the query-processing time on the server side and deliver rapid results to users. However, because the continuous *k*NN query involves numerous moving objects, object-by-object scanning considerably reduces the query-processing performance [13]. To overcome this challenge and improve query-processing efficiency, location indexing techniques have been developed and are deemed necessary.

Traditional database indexing sorts the data to store and access records effectively. However, these methods are not appropriate for frequently changing the data or location of moving objects due to the high cost of updates [14]. Moving object location indexing is a challenging research field, with proposed index structures based on spatial, time, or combined spatial and time indexing technologies. The research at present is still in the preliminary stage, both domestically and abroad [15].

The spatial index can be divided into categories according to the basic unit of index structure. Representative indexes include grid, tree-like, and hybrid indexes [16]. In [17], Yang et al. proposed an efficient distributed solution for *k*NN queries, which can handle larger amounts of moving-object data. The proposed solution included a new network-based index called BGI (block grid index). The BGI is an in-memory index based on a distributed hierarchical grid with a minimum and maximum predefined number of moving objects per block. Additionally, Park et al. [18] proposed a DGI (distributed grid index), which builds an index in the form of a hierarchical grid according to the location of objects. S-GRID (scalable grid) is also one of the very classic grid indexes that was proposed by Huang et al. [19] to deal

with the problem of road network indexing. Although these grid-based indexes pre-define the number of moving objects or layer the grid, as the moving objects and road networks change, the index cannot rapidly adapt or produce efficient query responses. This limits the efficiency of these indexes in large, dynamic, road network graphs.

In addition, the most common tree-like indexes are R-Tree, K-D Tree, Z-Tree, and Quad-Tree indexes. Among them, R-Tree and PMR Quad-Tree [20] indexes have a good performance for indexing traditional spatial objects. In order to further improve the query efficiency of kNN, Zhong et al. proposed a balanced search-tree G-Tree (grid tree) index structure in 2013 [21]. Based on G-Tree, the G\*-Tree index improves the query efficiency of G-Tree by dividing each node into two disjoint regions [22]. This optimization is based on a simple observation that, during querying, when a region is completely contained within the query range, other regions that do not intersect with this region can be ignored. Recently, Bareche et al. [23] proposed the Velocity SpatioTemporal indexing approach. The proposed structure is based on a selective velocity-partitioning method that reduces the update cost and improves the response time and query precision. Bilong Shen [24] of Tsinghua University proposed a new index: V-Tree. It can support a *k*-nearest neighbor search based on the road network. Second, it can support the dynamic updating of moving objects. The V-Tree index's structure is optimized further, based on G-Tree, to become the kNN query for moving objects. While these methods have shown a good performance in indexing traditional spatial objects, they may face several limitations when applied to the query of moving objects in large road networks. One limitation is that these methods may not efficiently handle the continuous updates of the location's information of moving objects, resulting in high update costs and query latencies. Another limitation is that they may not consider the road network's topology, which may result in inaccurate kNN queries and longer query times. Furthermore, these methods may not take into account the different speeds and directions of moving objects, leading to inefficient queries and inaccurate results. Correspondingly, the index proposed in this paper has an efficient update strategy and can cope with changes in large road networks with low maintenance costs.

There are two problems in the existing study. On the one hand is the question of applicability. Most of the existing methods are based on Euclidean distance constraints [25], and it is difficult to solve the problem of the continuous k-nearest neighbor query of moving objects based on road network constraints [26]; however, the distance between moving objects in real life is often determined by the length of the road network. Secondly, some methods are based on the fact that the query point is fixed [27], and only the position of the moving object changes, and then the continuous k-nearest neighbor incremental query algorithm of the moving object is designed based on this assumption. Once the position of the query point changes, the query result needs to be recalculated. Another aspect is the issue of efficiency. Although some methods can support the k-nearest neighbor query of moving objects based on the road network, when the road network scale is large, the number of query calculations significantly increases, resulting in reduced query efficiency, the poor real-time performance of query results, and insufficient practicability [28]. At times, the structure of the moving object index is complex, resulting in a large-scale update of the location of the moving object [29]; the update efficiency of the moving object's index structure is low, resulting in a decrease in the accuracy of the query result.

# 3. Methodology Overview and Problem Definition

In this section, we introduce the system overview and some relevant definitions for this study.

### 3.1. Methodology Overview

Unlike some existing methods that use tree-like indexes and create rules to optimize the search efficiency, we identified some issues with this approach. First, numerous leaf nodes cannot be efficiently pruned, which leads to numerous unnecessary vertices in these nodes.

Second, not all leaf nodes in the index are used for location-based services, and some of them have a minor effect, but still require significant costs for the construction and updates.

Based on the abovementioned problems, our scheme adopted the overall idea of parallel and balanced computing to build high-quality indexes. As shown in Figure 1, we first recursively partitioned the whole graph into multiple-grid subgraphs according to the quadtree structure, and then constructed new grid indices to take over these subgraphs. The moving object can determine the grid unit to which it belongs according to its position and compile it into the index of the corresponding grid unit. The granularity of meshing is an important factor that directly affects the performance of index building and updating behaviors. If the granularity is too large, there are too many nodes in the grid, and the function of dividing the units is lost; if the granularity is too small, the index needs to be frequently updated, which affects its overall performance.



Figure 1. Methodology Overview.

3.2. Problem Definition

**Definition 1.** Road Network. In this paper, we abstractly modeled the real road network as a directed weight graph,  $G = \langle V, E \rangle$ , each road was regarded as an edge of G, and the road endpoints were regarded as the corresponding vertices of G, where V is the set of nodes on the road and E is the set of edges on the road. A vertex on the road is  $v_i = (x_i, y_i) \in V$ , x is the abscissa of the vertex, y is the ordinate of the vertex; an edge on the road is  $e = (u, v, w) \in E$ , u is the road. The starting point of e is the end point of the road, and w is the length of the road. The road network used to illustrate the concepts in this paper is shown in Figure 2.



Figure 2. Diagram of road network.

**Definition 2.** *kNN Query. The NN query problem originated from the post office problem proposed by Knuth in 1973. The problem can be simply described as: given that set P in the NN -dimensional space contains n data points, find a data point p in set P such that the distance between p and the query point q is the closest. The kNN(k \geq 1) query is a generalized form of the NN query. In this paper, the k-nearest neighbor query is mainly performed on the moving objects in the road network environment, that is, on the road network G = \langle V, E \rangle, there is a group of moving objects* 

 $O_r(|O_r| > k)$ ; given a query point q and a positive integer k, find the k moving objects M(q) closest to the q-way network. In addition:

- |M(q)| = k;
- $Mq \subseteq O_r;$
- $\forall o \in O_r, o' \in O/M(q), Dist(q, o) \leq Dist(q, o')$

**Definition 3.** CkNN Query. In our semantics, moving objects move along the edges of the road network G and periodically report their positions with time interval  $\Delta t$ . The snapshots  $\Delta t$  of all objects are then updated at each time interval. Assuming that the latest snapshot of an object was created at time point  $t_s$  and all queries are made at time period  $(t_s, t_s + \Delta t]$ , the evaluation is first based on this snapshot. Subsequently, the kNN of each query is periodically updated based on each new snapshot of the object. We call queries following this semantics CkNN queries.

## 4. Dynamic V-Tree Double-Layer Grid Index

In this section, we first define the DVTG-Index and discuss how to construct it. Then, we describe the maintenance strategy, and conclude this section by discussing how to update the DVTG-Index.

# 4.1. Building the Index

To build the index, first we divided graph *G* corresponding to the global road network into d layers according to the quadtree structure, each layer was marked as  $L_i$  ( $i \in [1, d]$ ), and the  $L_1$  layer was the top layer. The area was divided into 4 grids of the same size, and each grid was divided into four grids at the  $L_2$  layer. According to this method, recursively divide until the moving objects contained in the smallest grid do not exceed  $\lambda$ , and the  $L_d$ layer divides the global road network into  $4^d$  grids. A grid unit of layer  $L_i$  is recorded as  $g_i^j$  ( $j \in [1, 4^j]$ ). As shown in Figure 3, each grid  $g_d^j$  of the  $L_d$  layer covers all the edges and vertices of graph *G* in this region. The divided subgraph is divided into a  $4^d$  branch tree, that is, the root corresponds to the whole graph and the leaf matches the smallest subgraph, that is, the underlying grid subgraph.



Figure 3. Diagram of road network division.

As shown in Figure 3, the second-layer grid index has 4 grid subgraphs, namely,  $g_2^1, g_2^2, g_2^3$ , and  $g_2^4$ , and the third-layer grid index has 16 grid subgraphs.

Then, we constructed the DVTG-Index, which replaces the structure of this subgraph and removes all the subgraphs covered by non-leaf nodes. The DVTG-Index only retains the complete topology of the original graph. Non-leaf nodes are considered virtual nodes and await materialization as they are required to provide indexing services. On the grid subgraph of each leaf node, the shortest distance between several vertices was calculated in parallel, offline, and used as the basic index material for indexing. Furthermore, we introduced two special types of vertices and some related concept definitions for each subgraph.

**Definition 4.** Active Vertex. If the moving object  $O_r$  is on the vertex  $v_a$  or the edge  $e_{b,a}$  and moves from vertex  $v_b$  to vertex  $v_a$ , then vertex  $v_a$  is defined as the active vertex and  $O_r$  is associated with  $v_a$ . Moving objects: active vertex  $v_a$  saves and maintains a collection of all moving objects associated with it, denoted as  $Q_a$ . The vertex set consisting of active vertices in grid  $g_i^j$  is denoted as  $A_i^j$ .

**Definition 5.** Boundary Vertex. If two vertices  $v_a$  and  $v_b$  on an edge  $e_{a,b}$  are given, they do not belong to any grid unit of the same layer  $L_i$  at the same time; then,  $v_a$  and  $v_b$  are both boundary vertices of layer *i*. The vertex set consisting of boundary vertices in grid  $g_i^j$  is denoted as  $\mathcal{B}_i^j$ .

**Definition 6.**  $Path(v_u, v_w)$ . In a connected path network, there are one or more paths from one vertex  $v_u$  to another vertex  $v_w$ , and the set of all paths is recorded as  $Path(v_u, v_w)$ . The elements in the collection are the edges in the path, which can be expressed as:

$$Path(v_u, v_w) = \{(e_{u,a}, \dots, e_{b,w}), \dots, (e_{u,x}, \dots, e_{y,w})\}.$$
(1)

**Definition 7.**  $Dist(v_u, v_w)$ . In a connected road network, there are one or more paths from one vertex  $v_u$  to another vertex  $v_w$ , and the sum of the distances of all paths is recorded as  $Dist(v_u, v_w)$ . The elements in the set are the sum of the distances of the edges in the path, which can be expressed as:

$$Dist(v_{u}, v_{w}) = \{(w_{u,a} + \dots + w_{h,w}), \dots, (w_{u,x} + \dots + w_{u,w})\}.$$
(2)

**Definition 8.**  $SPath(v_u, v_w)$ ,  $SDist(v_u, v_w)$ . In the path set  $Path(v_u, v_w)$  between two vertices,  $v_u$  and  $v_w$ , the path with the shortest side length and distance is called the shortest path, denoted as  $SPath(v_u, v_w)$ , and the distance of this path is denoted as  $SDist(v_u, v_w)$ .

Since any path from vertex  $v_u$  to vertex  $v_w$  outside the grid unit to which it belongs must pass through a certain boundary vertex x of the grid unit,  $Dist(v_u, v_w)$  is expressed; therefore, the shortest path of  $v_u$  to  $v_w$  can be expressed as:

$$SPath(v_u, v_w) = SPath(v_u, x) + SPath(x, v_w).$$
(3)

The shortest path distance can be expressed as:

$$SDist(v_u, v_w) = SDist(v_u, x) + SDist(x, v_w).$$
(4)

When there is an active vertex  $v_a$  in the grid, that is, there is a moving object  $p_r$  moving towards  $v_a$ , then  $p_r$  is in  $Q_a$ . Then, the shortest distance between another vertex  $v_b$  belonging to the same grid and moving object  $p_r$  indexed by  $v_a$  can be expressed as:

$$SDist(v_b, p_r) = SDist(v_b, v_a) + \delta(p_r, v_a).$$
(5)

In the underlying grid  $g_d^k$ , only the boundary and active vertices in the *k*th grid of its *d* level are included. Taking the road network in Figure 3 as an example, when  $\lambda = 2$ , the grid is divided into the second layer, and the division of the  $L_2$  layer graph and the settings of active and boundary vertices are shown in Figure 4.



Figure 4. Schematic diagram of active and boundary vertices.

As shown in Figure 4,  $v_1$ ,  $v_8$ , and  $v_{19}$  are active vertices,  $v_3$ ,  $v_4$ ,  $v_5$ ,  $v_7$ ,  $v_{12}$ ,  $v_{17}$ , and  $v_{18}$  are boundary vertices, and  $v_9$ ,  $v_{11}$ ,  $v_{16}$ , and  $v_{20}$  are both active and boundary vertices.

The DVTG-Index retains the grid subgraph of the leaf nodes; therefore, the grid g can be used to form a graph G, and each grid subgraph can be expressed as  $g = \langle V_g, E_g \rangle$ . The grid subgraphs in the DVTG-Index do not overlap each other, and all grid subgraphs contain all vertices belonging to the graph. Given the two vertices,  $v_a$  and  $v_b$ , belonging to the same grid, we used  $RelSDist(v_a, v_b)$  to represent the shortest distance between  $v_a$  and  $v_b$  within the grid, which was the relative shortest distance. Relatively,  $SDist(v_a, v_b)$  represents the shortest distance. It should be noted that the shortest distance expressed by RelSDist is only relative to the grid subgraph, not necessarily the shortest distance in the whole graph. Therefore,  $RelSDist(v_a, v_b)$  is not necessarily equal to  $SDist(v_a, v_b)$ ; it should satisfy:

$$RelSDist(v_a, v_b) \ge SDist(v_a, v_b).$$
 (6)

To make searching for grids more efficient, we indexed the shortest distance between the boundary and active vertexes in each grid. This allowed us to access the internal active vertex and its associated objects through any boundary vertex, which reduced the detection cost of the grid. We also used active nodes and boundary vertices to help identify the *k*NN. To further optimize the index structure, we introduced the concepts of vertex subgraph, vertex distance matrix, and vertex road network graph. The vertex subgraph only included active vertices, boundary vertices, and the shortest distance to a leaf node.

**Definition 9.** Vertex Subgraph. Let  $v_{i,a}^j$  and  $v_{i,b}^j$  be active and boundary vertices, and the active and boundary vertices sets in the grid  $g_i^j$  are marked as  $\mathcal{A}_i^j$  and  $\mathcal{B}_i^j$ ; the vertex subgraph is recorded as  $ChiG_i^j = \langle V_i^j, E_i^j, w_i^j \rangle$ , which satisfies:

- $V_i^j = \mathcal{A}_i^j \cup \mathcal{B}_i^j;$
- $E_i^j$  represents the edge between  $V_i^j$ , then  $e_{i,x,y}^j$  is  $v_{i,a}^j$  and  $v_{i,b}^j$  and other boundary vertices, that is,  $e_{i,x,y}^j = SPath(v_{i,x}^j, v_{i,b}^j), v_{i,x}^j \in V_i^j$ ;
- $w_i^j$  represents the distance of side  $E_i^j$ , that is,  $w_i^j = SDist(v_{i,a}^j, v_{i,b}^j) = min\{(w_{a,m} + \dots + w_{n,b}), \dots, (w_{a,u} + \dots + w_{w,b})\}.$

As shown in Figure 5,  $v_2$  is neither an active nor a boundary vertex. When searching inside the road network, we can prune such nodes; therefore, only active and boundary vertices are kept when constructing a vertex subgraph, while the edge in the graph is a virtual edge that represents the shortest route between two points and does not represent the actual meaning. The weight value of the edge is the distance of the shortest route.



Figure 5. Vertex subgraph diagram.

**Definition 10.** Vertex Distance Matrix. Each vertex subgraph  $ChiG_i^j$  needs to maintain a vertex distance matrix, saving the shortest distance  $SDist(v_a, v_b)$  between each active vertex and each boundary vertex. This distance matrix is recorded as  $DistM_i^j$ , which satisfies:

$$DistM_{i}^{j} = \begin{bmatrix} SDist(v_{i,a_{1}}^{j}, v_{i,b_{1}}^{j}) & SDist(v_{i,a_{1}}^{j}, v_{i,b_{2}}^{j}) & \cdots & SDist(v_{i,a_{1}}^{j}, v_{i,b_{m}}^{j}) \\ SDist(v_{i,a_{2}}^{j}, v_{i,b_{1}}^{j}) & SDist(v_{i,a_{2}}^{j}, v_{i,b_{2}}^{j}) & \cdots & SDist(v_{i,a_{1}}^{j}, v_{i,b_{m}}^{j}) \\ & \vdots & \ddots & \vdots \\ SDist(v_{i,a_{n-1}}^{j}, v_{i,b_{1}}^{j}) & SDist(v_{i,a_{n-1}}^{j}, v_{i,b_{2}}^{j}) & \cdots & SDist(v_{i,a_{n}}^{j}, v_{i,b_{m}}^{j}) \\ SDist(v_{i,a_{n}}^{j}, v_{i,b_{1}}^{j}) & SDist(v_{i,a_{n}}^{j}, v_{i,b_{2}}^{j}) & \cdots & SDist(v_{i,a_{n}}^{j}, v_{i,b_{m}}^{j}) \end{bmatrix}.$$
(7)

**Definition 11.** Vertex Elevated Graph. All sub-grids at the bottom layer can form a graph G. Similarly, all vertex subgraphs form a vertex elevated graph. Let  $ChiG_i^j$  be the vertex subgraph, and the vertex elevated graph is recorded as  $G_v = \left\{ ChiG_d^1, ChiG_d^2, \dots, ChiG_d^{4^d} \right\}$ .

As the construction of the vertex subgraph is completed, at this time, based on all the vertex subgraphs in the graph, the vertex elevated graph  $G_V$  is constructed, and  $G_V$  includes all active and boundary vertices in the vertex subgraph.

The constructed vertex elevated graph considerably reduces the number of vertices and edges, thus providing efficient services for subsequent neighbor queries. Figure 6 presents the pruning process for constructing a vertex elevated graph. Taking the road network graph in Figure 4 as an example, we set the threshold  $\eta$  to 2, divided the road network graph into 4 grid subgraphs, reconstructed the active and boundary vertices in each grid, and then performed pruning. As shown in Figure 4, the original road network graph has 21 vertices and 32 edges, and the Vertex Elevated Graph after the pruning strategy has only 14 vertices and 24 edges.



Figure 6. The comparison between the vertex subgraph and original road network graph.

We built vertex subgraphs for all underlying grids, which were then continuously maintained for the real-time reporting of the motion of moving objects, updating active vertices, boundary vertices, and the vertex distance matrix  $Dist M_i^j$ . However, in fact, we found that many vertex subgraphs may not be useful for a certain query; however, updating it incurs a certain maintenance cost. Therefore, after constructing the vertex subgraph for the underlying grid subgraph, we will continue to scale the relevant vertex subgraph

according to the movement and distribution of the moving objects to adjust the index granularities of different regions. The specific maintenance and update mechanisms are discussed in detail in the subsequent section.

# 4.2. Update Mechanism of the DVTG-Index

Since the *k*NN problem we discussed in this study was based on moving objects in the road network environment, we needed to update the DVTG-Index as the moving objects frequently change in the road network. The update was divided into two parts: the first part updates the position of the moving object, that is, it updates the moving object  $O_r$ ; the second part changes the distribution of the moving objects, active vertices in the grid, and the division granularities of moving objects. We need to further divide the grid according to the threshold, so as to build a better vertex subgraph.

### 4.2.1. Update of Moving Objects

The real-time update of large moving objects is a major feature in the urban road network environment, and it is also a difficulty in this research scenario. This paper introduced the concept of the active vertex; therefore, binding the moving object to the corresponding active vertex can efficiently handle the problem of updating the position of the moving object. The specific details of the update strategy are:

First, in the road network, the motion scenes of moving objects can be summarized into the following three situations:

- Add moving object: we added a new object  $O_r$  on edge  $e_{u,v} = (u, v, w)$ , and  $O_r$  moved towards vertex v, for example, a taxi driver just exited the car or just finished the last task.
- Delete a moving object: we deleted a moving object  $O_r$  from the edge  $e_{u,v} = (u, v, w)$ , for example, the driver finished a task and left work, or the car broke down and could no longer carry passengers.
- Update the moving object: the position of the moving object  $O_r$  changed, which was the most common situation. We further subdivided it into two cases according to the variation in moving objects. The first case was that  $O_r$  was still on the edge  $e_{u,v}$ ; however, the distance from the active vertex v changed. In this case, we only needed to change the  $\delta$  attribute in  $O_r = (t, (u, v), \delta)$ , and the time complexity was only O(1); the second case was that the side where  $O_r$  was located changed, for example,  $O_r$ moved from side (u, v) to side (v, w). At this time, we only needed to combine the operations of (1) and (2), delete  $O_r$  from edge u, and add  $O_r$  on edge (v, w), that is, the cross-edge movement of the moving object was realized.

From the analysis above, we only needed to consider the operations of adding and deleting moving objects.

Add moving object: add moving object  $O_r$  to edge  $e_{u,v} = (u, v, w)$ . At present, there are two cases according to the state of the vertex. (1) If v is an active vertex, you only need to add  $O_r$  to the moving object collection  $Q_v$  maintained by v; (2) if v is not an active vertex and  $Q_v$  is an empty set, we first mark v as an active vertex, and at the same time add  $O_r$  to the moving object collection  $Q_v$  maintained by v. Then, judge whether the number of moving objects in the vertex subgraph where v is currently located is within the threshold range. If it is within the range, the vertex subgraph does not need to be split or merged. Then, calculate the shortest distance from this point to other boundary vertices in the vertex subgraph and update the shortest distance matrix. If not within the threshold range, iteratively split or merge vertex subgraphs, and then calculate the shortest distance matrix of the new vertex subgraph.

Remove moving objects: remove  $O_r$  from edge  $e_{u,v} = (u, v, w)$ . At this point, two situations still need to be considered: (1) if the number of moving objects contained in the moving object set  $Q_v$  maintained by vertex v is more than 1, then it is only needed to delete  $O_r$  from  $Q_v$ ; (2) if there is only one moving object  $O_r$  in  $Q_v$ , then we also delete  $O_r$  from  $Q_v$  first, because  $Q_v$  is empty at this time, that is, no moving object moves to v,

then v becomes a normal vertex. Then, it is also judged whether the number of moving objects in the vertex subgraph where v is currently located is within the threshold range, and corresponding merging and splitting operations are performed.

## 4.2.2. Index Update

Updates to moving objects cause the number of active vertices within the vertex subgraph to change. At present, the distribution of moving objects in the city is mostly a Zipf distribution, not an even distribution; therefore, the number of active vertices in different vertex subgraphs is quite different, and the corresponding vertex distance matrix also requires higher calculation and space costs. In order to more reasonably balance the global computing load, the vertex subgraph needs to merge or split the subgraph according to the relationship between the number of active vertices of itself and the surrounding and set thresholds. In this study, the threshold value of number of active vertices  $v_a$  in the subgraph was set to  $\eta$ .

When adding a moving object, if the number of active vertices in the current vertex subgraph  $g_j^i$  is more than  $\eta$ , the subgraph g is split into four subgraphs, and furthermore, the four subgraphs of g are updated to calculate the corresponding shortest distance matrix. If the number of moving objects in a certain subgraph still meets the split condition, continue to recurse the abovementioned operations, layer by layer, until the split condition is not met.

When deleting a moving object, if the sum of the number of active vertices in the current vertex subgraph  $g_j^i$  and the number of active vertices in its three sibling vertex subgraphs is less than the specified threshold  $\eta$ , merge them into a new vertex subgraph  $g_{j'}^{i-1}$ . If the number of active vertices of the parent subgraph  $g_{j'}^{i-1}$  of  $g_j^i$  and its three sibling subgraphs is still less than  $\eta$ , continue to recursively operate layer by layer until the number of active vertices in the upper subgraph  $g_{j'}^i$  of  $g_j^d$  ( $1 \le i < d-1$ ) is not less than  $\eta$  or i = 1.  $g_{j'}^{i-1}$ . Save and maintain all the active vertices in the four subgraphs, update the border vertices, and recalculate the shortest path  $SPath(v_u, v_w)$  between any border and other vertices; save the shortest distance  $SDist(v_u, v_w)$  between vertices and the vertex distance matrix  $DistM_i^j$ .

When the road network map is large enough, there will be a high number of moving objects that need to be updated at the same time. Therefore, we used the method of scanning subgraphs to update all the moving objects in batches, and then updated the vertex subgraphs, which greatly improved the update efficiency of the DVTG-Index.

According to the description of the index update algorithm, if updating the moving object does not change the attributes of the vertices, the time complexity of updating is O(1). If the update operation changes the attributes of each vertex and changes the index structure, it needs to be discussed separately. Since the vertex subgraph needs to be recursively split or merged every time a moving object is added/deleted to balance the computational load, the time complexity of each adding/deleting a moving object is related to the recursive depth of the vertex subgraph. Therefore, it can be observed that the time complexity of the algorithm is  $O(\log n)$ . Because each operation may cause the recursion depth of the vertex subgraph to change, the time complexity can only be used as an approximation.

The index update algorithm can effectively balance the global computing load; however, it also needs to consider the impact of space complexity, because splitting subgraphs increases the space usage. We tested this in the experiments in the subsequent section.

With the update of the index of the change in the position of the moving object, taking the road network in Figure 7 as an example, when there are more moving objects in the grid subgraph in the lower left corner of the original road network diagram, as shown in Figure 5, the number of active vertices reaches three, which exceeds the threshold. According to the index update mechanism, the grid continues to be divided, and then corresponding pruning is performed to form a new vertex subgraph. The updated vertex elevated graph

has 17 vertices and 28 edges. Compared with the original road network graph, the number of the overall vertices and edges is still effectively reduced, and the number of active vertices in a single vertex subgraph is also more balanced, which reasonably allocates the computing cost in the subgraph and improves the overall data-processing capability.



Figure 7. The schematic diagram of index update instructions.

## 5. DVTG-CkNN Query Algorithm

We discussed the basic concept of the *k*NN query in the second section, and this section describes the process of the *k*NN query algorithm in detail. The *k*NN query problem that is usually studied is a single query, that is, given the position of the query point q, from graph G, find the *k* moving objects closest to q. However, in daily life, there are often multiple and continuous queries. In the same location, especially in the downtown area, there are often multiple query requests. If we regard each query request as a single query, it produce numerous repeated calculations, which will affect the efficiency of the *k*NN query. Therefore, we proposed the DVTG-C*k*NN algorithm based on the DVTG-Index. For each first query request, we used the DVTG-*k*NN algorithm to calculate the initial *k*NN of query q. Then, with the movement of the moving object, the *k*NN of the continuous query request q was updated round by round in the incremental search mode, and the search range of the initial query was saved, so that when the position was queried again, effective incremental calculations could be performed to improve the efficiency of the algorithm.

## 5.1. Query-Point Processing

Due to the randomness of the query points, we needed to preprocess the query points. The processing method is similar to the method used for processing moving objects in Section 4.2.1. We associated the location of query request q with the vertices in the road network.

In summary, the query point *q* query moving object problem in the road network was transformed into a vertex query problem in the road network.

### 5.2. The First k-Nearest Neighbor Query

According to the above points, we associated the query point with the vertices of the vertex subgraph, and the query between objects was converted into the query between vertices. In order to better explain the query algorithm, we first need to define some concepts.

**Definition 12.** *Vertex Queue S. We defined vertex queue S as:* 

$$S = \{q\} \cup \left\{ v \in V \middle| \begin{array}{c} v \text{ is a boundary or active vertex} \\ in the vertex subgraph where q resides} \right\}.$$
(8)

During the search process, the vertices in the processed vertex subgraph are continuously added to *S* until the algorithm terminates or the graph *G* is traversed. We can express this process as follows:

$$S_{t+1} = S_t \cup \{v \in V | v \text{ is a vertex in the processed subgraph at iteration } t\}.$$
 (9)

where  $S_{t+1}$  denotes the state of the vertex queue at iteration *t*.

Since graph G is a directed and weighted graph, all road networks can be connected through boundary vertices. The S queue represents the scope of the search map at present. During the search process, the vertices in the processed vertex subgraph are continuously added to S until the algorithm terminates or the graph G is traversed.

**Definition 13.** Query Queue  $F_q$ . The query queue  $F_q$  stores the *n* moving objects closest to the query point *q*, that is, the final result of the *k*-nearest neighbor query. In  $F_q$ , the road network distance between the moving object  $O_n$  that is farthest from *q* and *q* is denoted by  $\zeta$ . Therefore, we can mathematically represent  $F_q$  as follows:

$$F_q = \{O_i | i = 1, 2, \dots, n\}.$$
(10)

where  $O_i$  is the *i*-th moving object in the queue  $F_q$ . The distance between each moving object  $O_i$  and query point q is denoted by  $SDist(O_i, q)$ . Let  $\zeta$  be the maximum distance between any moving object  $O_i$  and q. Therefore, we can express  $\zeta$  as:

$$\zeta = \max(SDist(O_i, q)). \tag{11}$$

When the algorithm ends, the distances from other mobile objects  $O_n$  to the query point q are greater than  $\zeta$ , that is:

$$SDist(O_n',q) \ge \zeta, \forall O_n' \notin F_q.$$
 (12)

**Definition 14.** *Query Vertex Queue D. The query vertex queue D stores the vertices that have been determined to have the shortest distance to the query point. Queue D is initially created as an empty set. The vertices are added to D during the search process as their shortest distances are determined. Mathematically, we can define D as:* 

$$D = \{v | v \text{ is a vertice that has been determined to have the shortest distance to } q\}.$$
 (13)

Queue *D* is convenient for subsequent continuous queries, because the boundary vertices are not updated as the position of the moving object changes. Therefore, the border vertex data in the queue can still be used again in continuous queries, and the number of border vertices is higher than that of active vertices; therefore, the repeated traversal of vertices can be reduced.

Then, we detailed the steps of the first *k*-nearest neighbor query algorithm:

When the system received a *k*-nearest neighbor query request (q, k), it first associated query point q with its nearest vertex v; if v is not an active or boundary vertex of the vertex subgraph, it means that vertex v is not in the vertex set of the vertex subgraph, and the distance from vertex v to other vertices in the vertex subgraph is not calculated. Therefore, we needed to activate v as an active vertex first, add it to the vertex set of the vertex subgraph, update the index, and obtain the shortest distance between v and any boundary vertex in the vertex subgraph and the active vertex. If v itself is an active or boundary and active vertices in this vertex subgraph are then added to vertex sequence S.

Then, initialize query queue  $F_q$  to be empty, and traverse all vertices in S, find the vertex  $v_1$  closest to vertex v, and start processing vertex  $v_1$  (the first vertex to be processed is the query vertex v itself, because the weight of the road is a positive number and SDist(v, v) = 0; therefore, there is no need to consider the vertex itself). If  $v_1$  is the active vertex, add  $v_1$ 's moving object set  $Q_v$  to  $F_q$ , and the distance from the moving object  $O_n$  in the set to the query point q is  $SDist(v_1, v) + \delta$ ; if  $v_1$  is a border vertex, it is necessary to expand the search area and traverse all adjacent border vertices u of  $v_1$ . If u is in vertex queue S and  $SDist(u, v_1) + SDist(v_1, v) < SDist(u, v)$ , update  $SDist(u, v) = SDist(u, v_1) + SDist(v_1, v)$ . If u is not in vertex queue S or query vertex queue D, add u to S. At this point, the processing of vertex  $v_1$  is completed, and then  $v_1$  is deleted from S and added to query vertex queue D. Then, traverse all the nodes in S again, and continue to process the subsequent vertex closest to

query vertex v until the algorithm ends early or S is empty (if the algorithm ends early, it means that there are k or more moving objects in the query queue, and if S is empty, it means that all the connected graphs have been searched).

If the number of moving objects in  $F_q$  is greater than k, more than k moving objects have been found (because the number of moving objects contained in each active vertex is different, the number of moving objects saved in  $F_q$  will not be exactly k every time), then sort them according to the road network distance from the query point. Whenever the subsequent vertex  $v_n$  is processed, a judgment is made: if  $SDist(v, v_n) \ge \zeta$ , the algorithm ends early. Because the Dijkstra-style algorithm searches for vertices in the order of near to far, if  $SDist(v, v_n) > \zeta$ , the distance from the subsequent active vertex ucontaining the object moving to the query vertex SDist(v, u) must be greater than or equal to  $SDist(v, v_n)$ ; therefore, the distance between the object  $O_n$  moving towards u and the query point must be greater than or equal to SDist(v, u), and thus greater than  $\zeta$ . From this, it can be determined that k moving objects in  $F_q$  are the k moving objects closest to q. If  $SDist(v, v_n) < \zeta$  and the processed node is an active vertex, the moving object  $O_n$  in this point may still be closer to the query point than the moving object in  $F_q$ ; then, we need to compare  $O_n$  with the moving object in  $F_q$  and update it. The pseudocode of the first k-nearest neighbor algorithm is shown in Algorithm 1.

Algorithm 1. The first *k*-nearest neighbor query

Input: $q_k / /$ Find the <i>k</i> closest moving objects to the query point <i>q</i> Output: $F_q / /$ Returns the <i>k</i> -nearest neighbor query queue
function kNN(q, k)
Associate q with vertex v nearby
add v into $G_V$ ;
add v into S;
$F_a \leftarrow \varnothing, \zeta \leftarrow \infty;$
$D \leftarrow \emptyset;$
while $S \neq \varnothing$ do
$Dist \leftarrow \infty;$
for each vertex u in S do
if $SDist(v, u) < Dist$ then
$Dist \leftarrow SDist(u, v);$
$if F_a  = k$ and
$\zeta \leq Dist$ then
break;
if $u$ is in $A$ then
add $u.Q_u$ into $F_q$
$\zeta \leftarrow \min\{\zeta, \max\{SDist(O_n, v)\}\}$
else
<b>for</b> each vertex w in u's neighbors <b>do</b>
<i>if</i> $w \in S$ and $SDist(v, w) > SDist(v, u) + SDist(u, w)$ <i>then</i>
SDist(v, w) = SDist(v, u) + SDist(u, w)
else if $w \notin D$ then
add w into S
remove u from S
add u into D
return F <sub>a</sub>

5.3. Continuous k-Nearest Neighbor Query

Since the boundary vertices are only related to the road network itself, as long as the grid division does not change, the properties of the vertices as boundary vertices will not change. As the position of the moving object changes, only the active vertices are changed. Additionally, the number of boundary vertices is far greater than that of active vertices. Therefore, when query request *q* has already performed the first *k*-nearest neighbor query

operation, we can simplify the continuous *k*-nearest neighbor query based on the query vertex queue *D*. Perform the query from the range that was searched the previous time, and continue to expand it on the basis of this queue, instead of starting from scratch. The specific steps of the continuous *k*-nearest neighbor query algorithm are as follows.

We first associated query point q with the nearest vertex v, similar to the first k-nearest neighbor value. Then, traverse the vertices in query vertex queue D saved after the initial k-nearest neighbor query operation; if the vertex v in it was the active vertex at the time of the last query and is still the active vertex at this time, add the moving objects in the moving object sequence  $Q_v$  maintained by v to the new  $F_q$ . If v is no longer the active vertex, proceed to the subsequent node. If v is a boundary vertex, traverse the adjacent vertices of v to determine whether there is a new active vertex u or a boundary vertex that does not belong to D. If so, add it to S. If it is an active vertex, judge whether the active vertex is less than the distance from the last vertex in D to query vertex v, which is the furthest distance from the query point among the confirmed nodes. If it is less than, then update v's mobile object sequence  $Q_v$  to  $F_q$ , delete u from S, and insert it into D.

If the number of moving objects in  $F_q$  exceeds k, and the distance from the processed vertex in D to v is greater than the distance  $\zeta$  from the moving object farthest from the query point in  $F_q$  to the query point, the algorithm is terminated early. If the algorithm still cannot end after traversing all the nodes in D, continue to follow the operation of the initial k-nearest neighbor algorithm to traverse query range sequence S and continue to expand the search range. The pseudocode of the continuous k-nearest neighbor query algorithm is shown in Algorithm 2.

Algorithm 2 Continuous k-nearest neighbor query
Input: $O_r$ , $v$ // Find the k closest moving objects to the query point q
Output: F <sub>q</sub> / / Returns the k-hearest heighbor query queue
function CkNN(q, k)
Associate q with vertex v nearby
add v into G <sub>V</sub>
$D \leftarrow v.D$ , $S \leftarrow v.S$
$F_q \leftarrow arnothing$
for each vertex u in D or S do
<i>if</i> $u \notin A \cup B$ <i>at this time then</i>
remove u from D or S
$Dist \leftarrow SDist(v, u)$
for each vertex u in D do
if $u \in \mathcal{A}$ then
add $u.Q_u$ into $F_q$
$\zeta \leftarrow \min\{\zeta, \max\{Dist(p_r, q)\}\}$
else
for each vertex w in u's
neighbors do
if $w \in \mathcal{A}$ then
if $SDist(u, w) > Dist$ then
add w into S
else
insert w into D
if $ F_q  = k$ and $\zeta \leq D$ ist then
ргеак
return F <sub>a</sub>

5.4. Time and Space Complexity of the DVTG-CkNN Algorithm

5.4.1. Time Complexity

Subsequently, let us analyze the time complexity of the DVTG-*k*NN algorithm. The association of a query point with its nearest vertex is a constant-level time complexity.

The act of updating the vertex subgraph and deriving distances can be viewed as an operation for each active vertex with a time complexity of O(N), where N is the number of all vertices indexed at the top level. However, the time complexity of calculating the distance of each vertex to other vertices is  $O(M^2)$ ; however, since the shortest distance between vertices was obtained by distributed computing in our method, M should be the largest number of vertices in all vertex subgraphs.

Then, the time complexity of enqueuing the vertex is O(1). The time complexity of the operation of traversing the vertex queue is O(N), where N is the number of vertices in the vertex queue. The time complexity of finding vertex v is  $O(\log N)$ , and the minimum heap is used for searching.

For the processing of vertex v, if it is an active vertex, the time complexity of adding its moving object set to the query queue is O(1); if it is a border vertex, the time complexity of traversing all its adjacent border vertices is O(K), where K is the degree of each border vertex.

Therefore, the total time complexity is  $O(M^2 + N \log N + KP)$ , where *N* is the number of all vertices in the top-level index, *M* is the number of vertices at most in all vertex subgraphs, *K* is the degree of each border vertex, and *P* is the number of border vertices. However, in practice, the time complexity of the *k*NN algorithm also depends on the value of *k*. When *k* is small, the time complexity of the algorithm is lower; when *k* is large, the time complexity of the algorithm is not provide the time complexity of the algorithm is lower; when *k* is large, the time complexity of the algorithm is higher. Therefore, choosing an appropriate value for *k* has a great influence on the performance of the algorithm, which we tested and proved in subsequent experiments.

However, the time complexity of DVTG-CkNN is difficult to accurately evaluate because it depends on the number of active and boundary vertices close to the query point and the relationship between the vertices. Ideally, the number of active vertices is less than the number of boundary vertices, and the number of adjacent vertices of each boundary vertex is also small. In this case, the time complexity of the algorithm can be assumed to be O(k). However, if the number of active vertices is high and the number of border vertices adjacent vertices is also high, the time complexity may be high. Therefore, the exact time complexity of this algorithm cannot be provided, and we tested it in the experiments in the subsequent section.

### 5.4.2. Space Complexity

Based on the algorithm's description, the space complexity of the DVTG-CkNN algorithm can be analyzed as follows:

The space required to store the vertex subgraph, including active, adjacent, and boundary vertices, can be considered as O(|V|), where |V| is the total number of vertices in the subgraph.

The space required to store the moving object sets of active vertices, including the query queue  $F_q$ , can be considered as O(|O|), where |O| is the total number of moving objects in the subgraph.

The space required to store vertex sequence *S* and query vertex queue *D* can also be considered as O(|V|). In addition, the space required to store the shortest distances between vertices, which are updated during the search, can be considered as  $O(|V|^2)$ .

Therefore, the total space complexity of the DVTG-CkNN algorithm can be approxi-

mated as  $O(|V|^2 + |O|)$ .

# 6. Experiment Analysis

In this section, we experimentally evaluated the performance of DVTG-Index and presented the performance of our proposed DVTG-CkNN algorithm when dealing with the *k*NN and continuous *k*NN queries. To fully evaluate the performance of our method, we chose three advanced methods, *G*\*-Tree, V-Tree, and SILC, as benchmark algorithms to compare to our method.

# 6.1. Experiment Settings

This experiment was based on the Java language program, and the IDE program used IDEA. The computer software/hardware configuration is shown in Table 1.

Table 1. The computer software/hardware configuration.

Software and Hardware	Version/Model	
Operating system	Windows 10 (1902)	
CPU	Intel Core i7-7700 HQ	
Memory	8 GB	
Hard disk	1 TB	
Java	Jdk1.8.2	

Additionally, we validated the performance of our query model by simulating a set of objects moving on a real road network. Therefore, we needed three types of datasets that were real road network, moving-object, and query-request datasets. This experiment selected the following three real-world road networks for the experiments: BJ (Beijing), NY (New York), COL (Colorado), and NW (Northwest USA) [30]. These networks have been widely used in previous studies, and the number of nodes in the road network ranges from hundreds of thousands to millions. The specific information is shown in Table 2.

Table 2. Experimental software and hardware configuration table.

Dataset	Name	Number of Vertices	Number of Sides
BJ	Beijing	188,229	436,648
NY	New York City	264,346	733,846
COL	Colorado	435,666	1,056,066
NW	Northwest USA	1,207,945	2,840,208

For the BJ road network set, we used a real-world dataset, BJ T-drive, containing actual taxi trajectories in Beijing from February 2 to August 2008, which included nearly 18 million GPS trajectories for 10,350 taxis. We used existing methods to map the locations of moving objects and queries onto road networks [31].

For the other three road network sets, we used the moving-object simulation generation software invented by Thomas Brinkoff [32] to realize the initial distribution and update the movement and query requests of moving objects in the road network. We temporarily set the total number of moving objects as one percent of the number of vertices, and we use the following three strategies to simulate the starting position and subsequent trajectories of the moving objects: RD (random distribution), ND (normal distribution), and ZD (Zipf distribution).

### 6.2. Baseline Algorithm

We compared the DVTG-Index with three benchmark methods: V-Tree, G\*-Tree, and SILC. These three methods were implemented in this experiment; the implementation of the first two methods was provided by their authors, and the implementation of SILC as completed by us. This is how it worked: we started by building an index structure for the (static) road network. Then, we created a reference index, such as the DVTG-Index, which saves a list of moving objects from the vertex to the vertex. Similar to our approach, if a vertex has some moving objects, SILC calls it a POI. Because POIs may contain multiple moving objects, POIs need to build a list of moving objects. Additionally, when looking for *k*NN moving objects, the algorithm will stop. SILC uses a best-first approach to compute *k*NN results. We also used this index to keep POIs, calculate top-*k*NN POIs, and add moving objects of interest points to the result set.

## 6.3. DVTG-Index Experiment Situation

In this section, we conducted experiments by changing the threshold  $\eta$  of the number of moving objects in the parameter grid; we conducted performance evaluations from two aspects: the time and space overhead of the index construction and the time and space overhead of the index update. In order to exclude the influence of other factors, the results of this experiment are the average values obtained after repeated experiments.

First, we evaluates the characteristics of the DVTG-Index, built a static DVTG-Index, V-Tree, G\*-Tree, and SILC on the datasets BJ, NY, COL, and FLA, and compared the time cost  $t_c$  and space cost of building the index  $m_c$ , as shown in Figure 8. Among them, the road network set BJ randomly selected the real distribution of taxis at 8:00, 12:00, and 17:00 on February 2, and the road network set NY used three different distribution strategies to set the starting position of the moving object.



**Figure 8.** Time and space costs of the DVTG-Index. (a) Time costs on BJ; (b) Space costs on BJ; (c) Time costs on NY, COL and NW; (d) Space costs on NY, COL and NW.

Among them, the construction time of the DVTG-Index was composed of the network construction time, vertex distance matrix calculation time, and vertex elevated graph construction time; construction space was composed of the road network set, boundary vertices and active vertex sets, vertex distance matrix, and vertex elevated graph. As shown in Figure 8a,b, 8:00 and 17:00 in Beijing are the morning and evening peak periods, respectively; therefore, there are more moving objects and increased time and space factors for the index construction. As shown in Figure 8c,d, the number of vertices and edges of the road network sets NY, COL, and NW increases sequentially. The constructed DVTG-Index needs to construct more vertex subgraphs and calculate more vertex distance matrices at the same time, so that the memory space occupied by the index increases sequentially and the construction overhead time also increases. In addition, different distribution strategies of moving objects also have a certain impact on time and space costs. Under the three distribution strategies of RD, ND, and ZD, the time and space costs of the constructed DVTG-Index increase in turn. With an uneven distribution, the indexed grid requires more merge-up and split-down operations, thus requiring more time and space overhead.

When the index is constructed, the number of grid divisions determines the number of grids in the initial index. According to the above factors, if the grid is divided l times, the number of grids is  $4^l$ . The number of grids determines the size of the grid area and determines how many nodes and moving objects are included in the grid; therefore, we changed the number of grid divisions l to evaluate the time and space overhead of the DVTG-Index on each dataset. To exclude the influence of other factors, in this experiment, we adopted the ZD distribution strategy, and the data shown in the chart are the average of the results of multiple experiments.

As shown in Figure 9, it can be observed from the experimental results that there is no positive or negative correlation between the construction time and space costs and the number of grid divisions *l*. As the number of grid divisions gradually increases, the time and space values overhead first decrease and then increase. This is because when the number of grid divisions is low, the number of grids is low, the area of a single index grid is large, and the number of active vertices in a single index grid is high, which makes the calculation time and space overhead of the shortest distance matrix in the grid larger. Due

to the small number of index layers, the calculation time and space overhead of the shortest distance matrix account for a large proportion of the total cost; therefore, as the grid is split, the total time and space costs of the index construction are significantly reduced. However, when the number of index layers gradually increases to a certain number, the time to split the index grid down gradually increases, and the number of shortest-distance matrices increases, which also makes its space cost increase sharply; therefore, the total time and space overhead of the index construction starts to increase as the number of index layers increases. In addition, we found that the experimental results of the time and space overhead of building indexes have similar trends in different datasets; therefore, we can use the NY dataset to evaluate the index performances in subsequent experiments.



Figure 9. Construction time and space costs overhead of DVTG-Index w.r.t. l. (a) Construction time and space costs on BJ; (b) Construction time and space costs on NY; (c) Construction time and space costs on COL; (d) Construction time and space costs on NW.

Then, we conducted comparative experiments for V-Tree, G\*-Tree, SILC, and the DVTG-Index.

As shown in Figure 10, it can be seen from the results in the figure that the construction time for V-Tree is significantly longer than that of the other indexes. This is because V-Tree must perform the Floyd algorithm for all subgraphs to calculate the distance matrix. Although the subgraphs in the upper layer only save the boundary points of the lower layer, the number of calculations is still very high. However, the DVTG-Index iteratively merges some useless vertex subgraphs upwards, reducing the number of boundary points and greatly optimizing the index construction. In addition, with the increase in the road network, the advantages of our method are more obvious.



Figure 10. Comparison of construction time and space costs between baseline algorithms and DVTG-Index. (a) Construction time and space costs on BJ; (b) Construction time and space costs on NY; (c) Construction time and space costs on COL; (d) Construction time and space costs on NW.

The real-time change in moving objects is one of the difficulties in this study. Our index associated the moving objects with the vertices of the road network index and mapped the changes in the moving objects to the update of the index, which better handled this difficulty. When the moving object started to move on the road network, the index update performance also directly affected the performance of subsequent k-nearest neighbor queries. The number of moving objects and the index grid threshold affect the update performance of the index. Therefore, in this section, we evaluated the impact of various parameters on the performance of the DVTG-Index update.



First, we evaluated the impact of the number of moving objects on the index update time. In the BJ dataset, select 10%, 30%, 50%, 70%, and 100% of all taxis as the number of moving objects. In datasets NY, COL, and NW, 0.5%, 1.5%, 2.5%, 3.5%, and 5% of the number of vertices are generated as the number of moving objects, respectively. Among them, the index grid threshold  $\eta$  is set to 8.

As shown in Figure 11, it can be seen from the experimental results that as the number of moving objects increases, the updated overhead time also increases. The result is consistent with the principle of our design; the increase in the number of moving objects leads to the more frequent splitting and merging of the index grid during the update process, and the calculation time of the vertex distance matrix also increases accordingly.



**Figure 11.** Time cost of updating DVTG-Index w.r.t. the number of moving objects. (**a**) Time cost on BJ; (**b**) Time cost on NY, COL and NW.

Then, we evaluated the impact of index grid threshold  $\eta$  on the index update time. Among them, in the BJ dataset, the taxi trajectory between 8:00–9:00 on February 2 was obtained, and in datasets NY, COL, and NW, 5% of the number of vertices was used as the number of moving objects.

As shown in Figure 12, it can be seen from the experimental results that there is no positive or negative correlation between the index update time cost and index grid threshold  $\eta$ . When the threshold is relatively small, as the moving objects change, the index grid is more likely to be split down or merged up until the number of moving objects in the grid meets the threshold range, which takes a long time to update the index. As the threshold gradually increases, the number of moving objects that can be accommodated in the grid increases, and the time for grid scaling decreases so the total updated overhead time decreases. However, when the threshold gradually increases to a certain value, the number of active vertices in a single index grid is also very high, which makes the calculation time of the vertex distance matrix in the grid increase sharply. This makes the total time cost of the index update gradually increase with the increase in the threshold. In addition, the results of different datasets show that there are corresponding optimal thresholds in different road networks to minimize the update time cost, and the optimal value of the threshold increases with the increase in the dataset.

Finally, we conducted comparative experiments on the index update performance of V-Tree, G\*-Tree, SILC, and the DVTG-Index. Dataset BJ takes 100% of taxis as moving objects, 5% of the number of generated vertices in datasets NY, COL, and NW as moving objects, and ZD as the moving-object distribution strategy.



Figure 12. Index update time of DVTG-Index w.r.t. η.

As depicted in Figure 13, the empirical results demonstrate that the DVTG-Index exhibits the lowest and a relatively stable update time cost. This observed performance improvement of the DVTG-Index over benchmark algorithms is attributed to the reduction in the number of updated objects by associating moving objects with active vertices, thereby effectively reducing maintenance costs. As the size of the road network dataset increases, the comparative advantage of the DVTG-Index becomes increasingly pronounced. Conversely, the update of V-Tree and SILC requires a high time cost because, in the update process, they need to frequently traverse all the boundary points in the subgraph to search for the nearest interest point of the vertex.



Figure 13. Comparison of update time between baseline algorithms and DVTG-Index.

### 6.4. Experimental Situation of the DVTG-CkNN Algorithm

The main aim of this paper was to solve the *Ck*NN query problem; therefore, this section is the most important part of our experiment. To evaluate the *Ck*NN query performance of our model, we compared our model with state-of-the-art methods. We evaluated our algorithm by varying the value of *k*, the number of moving objects, the strategy of moving-object distributions, and the number of consecutive queries.

First, we evaluated the effect of varying the value of *k* on the performance of our algorithm and other benchmark algorithms for *k*NN queries for datasets BJ, NY, COL, and NW. We changed the value of *k* to 1, 3, 5, 10, and 20, respectively, and recorded the overhead time required for related queries. Other experimental parameters presented continuous queries as 3, 10% of all taxis as moving objects in the BJ dataset, 2.5% of the number of vertices as the number of moving objects in the NY, COL and NW datasets, and the moving objects used the ZD distribution strategy.

As shown in Figure 14, obviously, as the value of k increases, the query times of all algorithms increase; the average continuous query time of DVTG-CkNN is the shortest among the three algorithms. Taking the BJ dataset as an example, for k = 5, the average single query time of DVTG-CkNN is about 79  $\mu$ s, while the average single query time based on the G\*-Tree is about 259  $\mu$ s and the average query time based on V-Tree is about 1132  $\mu$ s. It can be observed that the larger the value of k, the better the performance effect of the DVTG-CkNN algorithm than other algorithms. This is because the number of moving



objects in our index subgrid can be kept within a dynamic balance range, which reduces the complexity required to expand the search range and shortens the algorithm query time.

**Figure 14.** Comparison of query times between baseline algorithms and DVTG-*Ck*NN w.r.t. *k*. (a) Query times on BJ; (b) Query times on NY; (c) Query times on COL; (d) Query times on NW.

Then, we evaluated the impact of the number of moving objects on the performance of *k*NN queries on BJ and COL datasets. In the BJ dataset, we selected 10%, 30%, 50%, 70%, and 100% of all taxis as moving objects, and in the COL dataset, we, respectively, generated 0.5%, 1.5%, 2.5%, 3.5%, and 5% of the number of vertices as the number of moving objects and set other experimental parameters k = 5, and the distribution strategy of moving objects adopted the ZD distribution.

As shown in Figure 15, we can draw the following conclusions. Firstly, as the number of moving objects increases, the average query overhead time of the three algorithms decreases. This is because the increase in the number of moving objects increases the density of the moving objects in the search grid and reduces the range to be searched. Secondly, the average continuous query time of DVTG-CkNN is the shortest among the three algorithms; even in the case of the high number of moving objects, DVTG-CkNN still has a great advantage. As previously designed, the DVTG-CkNN algorithm is based on the DVTG-Index, which divides the large road network into grids and dynamically merges or splits them according to the number of active vertices in the grid. This keeps the moving objects in different grids in a relatively balanced range, which can significantly improve the efficiency of querying moving objects in large road networks.



**Figure 15.** Comparison of query times between baseline algorithms and DVTG-*Ck*NN w.r.t. the number of moving objects. (**a**) Query times on BJ; (**b**) Query times on NY; (**c**) Query times on COL; (**d**) Query times on NW.

Then, we evaluated the impact of different distribution strategies of moving objects on the performance of the *k*NN query using three distribution strategies to generate the number of moving objects for datasets BJ and NW, respectively. Experiment with other parameters k = 5, and the number of moving objects is 2.5% of the number of vertices in the dataset.

As shown in Figure 16, we can draw the following conclusions. First, under the three distribution strategies of RD, ND, and ZD, the overhead times of the three query algorithms increases in turn. The uneven distribution of moving objects under the three distribution strategies gradually increases, and the search range gradually increases, requiring more overhead time. Secondly, the average single query time and average continuous query time of DVTG-CkNN are the shortest among the three algorithms. Especially under the ZD

distribution strategy, the performance advantage of DVTG-CkNN is more obvious than that of the benchmark algorithm. This is because the DVTG-CkNN algorithm is based on the DVTG-Index, which is designed for unevenly distributed dynamic road networks, and balances the uneven distribution of moving objects by dynamically dividing the grid.



**Figure 16.** Comparison of query times between baseline algorithms and DVTG-*Ck*NN under the three distribution strategies. (**a**) Query times on BJ; (**b**) Query times on NY; (**c**) Query times on COL; (**d**) Query times on NW.

Finally, we evaluated the impact of different consecutive query times on the performance of *k*NN queries. We set the number of continuous queries as 1, 3, 5, 10, and 20, and the moving objects used the ZD distribution strategy. Other experimental parameters were k = 5, 10% of all taxis as moving objects in the BJ dataset, and 2.5% of the number of vertices as the number of moving objects in the NY, COL, and NW datasets, and the moving objects used the ZD distribution strategy.

As shown in Figure 17, the average continuous query time of DVTG-CkNN is the shortest among the three algorithms. Especially after the number of continuous queries gradually increases, the performance advantage of DVTG-CkNN compared with the base-line algorithm is more obvious. This is because the DVTG-CkNN algorithm provides traversal for subsequent continuous queries through the query vertex queue. The boundary vertices in the queue are not updated as the position of the moving object changes. Therefore, the border vertex data in the queue can still be used again in continuous queries, and the number of border vertices is greater than that of active vertices; therefore, the repeated traversal of vertices can be reduced.



**Figure 17.** Comparison of query times between baseline algorithms and DVTG-*Ck*NN w.r.t. the number of continuous queries. (a) Query times on BJ; (b) Query times on NY; (c) Query times on COL; (d) Query times on NW.

#### 7. Conclusions

This paper studied the problem of the continuous *k*-nearest neighbor query for moving objects in a road network, and proposed the DVTG-Index, a dynamic multi-layer grid index, which can index moving objects in the road network and update them in real time. Secondly, this paper provided a continuous *k*-nearest neighbor query method for moving objects in a road network based on the DVTG-Index. DVTG-C*k*NN can perform efficient and continuous *k*-nearest neighbor queries, and the query efficiency reaches the microsecond level, which is obviously better than the mainstream algorithm used at present for solving this problem. Finally, we conducted experiments on a real road network dataset. The experimental results show that the DVTG-Index and DVTG-CkNN proposed in this chapter present better performance. Especially in the face of large road network maps and uneven distributions of moving objects, it has more rapid response capabilities. We also know that our method presented limitations and deficiencies. The *k*-nearest neighbor query method proposed in this paper is based on the optimal condition of single query processing. However, when the actual platform processed query requests, the sum of all individual optimal query processes may not be the optimal processing method for the overall situation. In the following stage of the research, we will focus on the *k*-nearest neighbor query algorithm with globally optimal query processing methods, which is of great significance to improve the responsiveness of the platform, and it also provides a good direction for the follow-up research. At the same time, in real life, road conditions must be considered. The weight value of the road network should not only be the length of the road, but should also be added to the road traffic conditions, and the road traffic cost should be considered as the weight value. Therefore, the future work can consider optimizing the road network model and enhancing the practicability of the road network-based *k*-nearest neighbor query algorithm.

**Author Contributions:** Conceptualization, K.T. and Z.D.; Methodology, K.T.; Software, K.T.; Validation, K.T., Z.D. and W.S.; Formal analysis, K.T.; Investigation, K.T.; Resources, Z.D.; Data curation, K.T.; Writing—original draft, K.T.; Writing—review & editing, Z.D.; Visualization, K.T.; Supervision, Z.G.; Project administration, Z.G.; Funding acquisition, Z.G. All authors have read and agreed to the published version of the manuscript.

**Funding:** This research was funded by Shanghai Municipal Science and Technology Major Project (No. 2021SHZDZX0103), the Science and Technology Commission of Shanghai Municipality (No. 19511132000), and the Department of Science and Technology of Guangdong Province (Grant No. 2019A1515110352).

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

## References

- Liao, Z.; Yang, Z.; Huang, P.; Pang, N.; Zhao, X. Multi-Model Fusion-Based Hierarchical Extraction for Chinese Epidemic Event. Data Sci. Eng. 2023, 8, 73–83. [CrossRef]
- Huang, H.; Gartner, G.; Krisp, J.M.; Raubal, M.; Van De Weghe, N. Location based services: Ongoing evolution and research agenda. J. Locat. Based Serv. 2018, 12, 63–93. [CrossRef]
- Gu, Y.; Liu, G.; Qi, J.; Xu, H.; Yu, G.; Zhang, R. The moving *k* diversified nearest neighbor query. *IEEE Trans. Knowl. Data Eng.* 2016, 28, 2778–2792. [CrossRef]
- Zhong, W.; Chen, C. REMIX: Efficient range query for LSM-trees. In Proceedings of the 19th USENIX Conference on File and Storage Technologies, Santa Clara, CA, USA, 23–25 February 2021.
- Li, C.; Gu, Y.; Qi, J.; Yu, G.; Zhang, R.; Yi, W. Processing moving k nn queries using influential neighbor sets. *Proc. VLDB Endow.* 2014, *8*, 113–124. [CrossRef]
- Wang, G.; Lei, Y.; Zhang, Z.; Peng, C. A Communication Efficient ADMM-based Distributed Algorithm Using Two-Dimensional Torus Grouping AllReduce. *Data Sci. Eng.* 2023, *8*, 61–72. [CrossRef]
- Kumar, B.K.; Priya, R.H.; Lakshmi, M.D.; Asha, S.; Mercy, J.J. Enhancing the Privacy and Content-Protection using Location Based Queries. J. Crit. Rev. 2022, 4, 306–313.
- 8. Yin, Q.; Wang, J.; Du, S.; Leng, J.; Li, J.; Hong, Y.; Zhang, F.; Chai, Y.; Zhang, X.; Zhao, X.; et al. An Adaptive Elastic Multi-model Big Data Analysis and Information Extraction System. *Data Sci. Eng.* **2022**, *7*, 328–338. [CrossRef]
- 9. Zhong, H.; He, W.; Cui, L.; Liu, L.; Yan, Z.; Zhao, K. Joint Attention Networks with Inherent and Contextual Preference-Awareness for Successive POI Recommendation. *Data Sci. Eng.* 2022, 7, 370–382. [CrossRef]
- 10. Tang, C.; Wan, J.; Xie, C. Fencekv: Enabling efficient range query for key-value separation. *IEEE Trans. Parallel Distrib. Syst.* 2022, 33, 3375–3386. [CrossRef]
- Jiang, T.; Zhang, B.; Lin, D.; Gao, Y.; Li, Q. Efficient parallel processing of high-dimensional spatial k NN queries. *Soft Comput.* 2022, 26, 12291–12316. [CrossRef]
- 12. Arshad, S.; Hu, S.; Ashraf, B.N. Zipf's law and city size distribution: A survey of the literature and future research agenda. *Phys. A Stat. Mech. Its Appl.* **2018**, 492, 75–92. [CrossRef]

- 13. Nutanong, S.; Zhang, R.; Tanin, E.; Kulik, L. Analysis and evaluation of v\*-k nn: An efficient algorithm for moving k nn queries. *VLDB J.* **2010**, *19*, 307–332. [CrossRef]
- 14. Wei, H.; Chen, Y.; Tan, J.; Wang, T. Sambot: A self-assembly modular robot system. *IEEE/ASME Trans. Mechatron.* **2010**, *16*, 745–757. [CrossRef]
- 15. Elmasri, R.; Wuu, G.T.; Kim, Y.J. The time index—An access structure for temporal data. In Proceedings of the Sixteenth International Conference on Very Large Databases, Brisbane, Australia, 13–16 August 1990; pp. 1–12.
- Huang, X.; Jensen, C.S.; Šaltenis, S. Multiple k nearest neighbor query processing in spatial network databases. In Proceedings of the Advances in Databases and Information Systems: 10th East European Conference, ADBIS 2006, Thessaloniki, Greece, 3–7 September 2006; Springer: Berlin/Heidelberg, Germany, 2006; pp. 266–281.
- 17. Tao, Y.; Papadias, D.; Sun, J. The tpr\*-tree: An optimized spatio-temporal access method for predictive queries. In Proceedings of the 2003 VLDB Conference, Berlin, Germany, 9–12 September 2003; Elsevier: Amsterdam, The Netherlands, 2003; pp. 790–801.
- 18. Goh, C.H.; Lu, H.; Ooi, B.C.; Tan, K.-L. Indexing temporal data using existing b+-trees. *Data Knowl. Eng.* **1996**, *18*, 147–165. [CrossRef]
- 19. Lum, V.; Dadam, P.; Erbe, R.; Günauer, J.; Pistor, P.; Walch, G.; Werner, H.; Woodfill, J. Designing dbms support for the temporal dimension. *ACM Sigmod Rec.* **1984**, *14*, 115–130.71. [CrossRef]
- Rotem, D.; Segev, A. Physical organization of temporal data. In Proceedings of the 1987 IEEE Third International Conference on Data Engineering, Los Angeles, CA, USA, 3–5 February 1987; IEEE: New York, NY, USA, 1987; pp. 547–553.
- 21. Lomet, D.; Salzberg, B. Access methods for multiversion data. ACM SIG-MOD Rec. 1989, 18, 315–324. [CrossRef]
- Kolovson, C.; Stonebraker, M. Indexing techniques for historical databases. In Proceedings of the Fifth International Conference on Data Engineering, Los Angeles, CA, USA, 6–10 February 1989; IEEE Computer Society: Washington, DC, USA, 1989; pp. 127–128.
- 23. Bareche, I.; Xia, Y. A Distributed Hybrid Indexing for Continuous KNN Query Processing over Moving Objects. *ISPRS Int. J. Geo-Inf.* **2022**, *11*, 264. [CrossRef]
- 24. Huang, C.; Su, B.; Lin, T.; Huang, Y. Downlink SCMA codebook design with low error rate by maximizing minimum Euclidean distance of superimposed codewords. *IEEE Trans. Veh. Technol.* **2022**, *71*, 5231–5245. [CrossRef]
- Yang, H.; Cai, J.; Zhu, M.; Liu, C.; Wang, Y. Traffic-informed multi-camera sensing (TIMS) system based on vehicle re-identification. IEEE Trans. Intell. Transp. Syst. 2022, 23, 17189–17200. [CrossRef]
- 26. Akyürek, E.; Bolukbasi, T.; Liu, F.; Xiong, B.; Tenney, I.; Andreas, J.; Guu, K. Tracing knowledge in language models back to the training data. *arXiv* 2022, arXiv:2205.11482.
- Li, J.; Ni, C.; He, D.; Li, L.; Xia, X.; Zhou, X. Efficient k NN query for moving objects on time-dependent road networks. *VLDB J.* 2022, 1–20. [CrossRef]
- Duan, R.; Deng, H.; Tian, M.; Deng, Y.; Lin, J. SODA: A large-scale open site object detection dataset for deep learning in construction. *Autom. Constr.* 2022, 142, 104499. [CrossRef]
- 29. Segev, A.; Gunadhi, H. *Event-Join Optimization in Temporal Relational Databases*; Lawrence Berkeley National Lab. (LBNL): Berkeley, CA, USA, 1989.
- 30. Maruta, A.; Kato, M.P. Intent-Aware Data Visualization Recommendation. Data Sci. Eng. 2022, 7, 301–315. [CrossRef]
- Yu, H.; Guo, X.; Luo, X.; Bian, W.; Zhang, T. Construct Trip Graphs by Using Taxi Trajectory Data. Data Sci. Eng. 2023, 8, 1–22. [CrossRef]
- 32. Brinkhoff, T. A framework for generating network-based moving objects. Geoinformatica 2002, 6, 153–180. [CrossRef]

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.